Author's response to reviews

Title: Participatory Testing and Reporting in an Environmental-Justice Community of Worcester, Massachusetts: A Pilot Project

Authors:

Timothy J Downs (tdowns@clarku.edu)
Laurie Ross (lross@clarku.edu)
Mucciarone Danielle (danielle.mucciarone@gmail.com)
Maria Camila Calvache (camilacalvache@gmail.com)
Octavia Taylor (otaylor@clarku.edu)
Robert Goble (rgoble@clarku.edu)

Version: 5 Date: 23 June 2010

Author's response to reviews:

Cover letter for MS1054081311354521

Point-by-point:
• In the Abstract please insert a line break between the sections. DONE
• Please insert a page break between major sections of the manuscript (abstract, main text, references, tables. DONE
• We generally discourage the use of abbreviations in the abstract, as they make the reading much more difficult for colleagues, who scan PubMed or similar services. Please reword to this effect. DONE
• Please delete the subheadings from the Introduction and Results sections. DONE
• Please delete reference in the main text in regard to the placement of tables and figures. DONE
• Please void using abbreviations in subheadings. Although we occasionally allow subheadings in the Discussion section, they seem not to be essential, and the subheading level is unclear. DONE
• In the Authors' contributions, please state that 'All authors read and approved the final version of this manuscript.' DONE
• The references are not in accordance with our instructions, please correct. Also, please note problems with ref 57. Carefully REVIEWED AND REVISED TO CONFORM TO INSTRUCTIONS. NOT SURE WHAT THE PROBLEM WITH [57] WAS (Shendrikar and Steinmetz 2003?) WE CHECKED IT'S ACCURACY – BUT WE DID FIND A REPEATED REFERENCE [59] Jacobs et al. 2002 AND REMOVED IT, renumbering accordingly.

• Please also highlight (with 'tracked changes'/coloured/underlines/highlighted text) all changes made when revising the manuscript to make it easier for the
Editors to give you a prompt decision on your manuscript. DONE.
THANK YOU.